Author's response to reviews

Title: The influence of population characteristics on variation in general practice based morbidity estimations

Authors:

Catharina van den Dungen (karin.van.den.dungen@rivm.nl)
Nancy Hoeymans (nancy.hoeymans@rivm.nl)
Hendriek C. Boshuizen (hendriek.boshuizen@rivm.nl)
Marjan van den Akker (marjan.vandenakker@maastrichtuniversity.nl)
Marion C.J. Biermans (m.biermans@elg.umcn.nl)
Kees van Boven (cvboven@hetnet.nl)
Henk J. Brouwer (h.j.brouwer@amc.uva.nl)
Robert A. Verheij (r.verheij@nivel.nl)
Margot W.M. de Waal (m.w.m.de_waal@lumc.nl)
Francois G. Schellevis (f.schellevis@nivel.nl)
Gert P. Westert (g.westert@iq.umcn.nl)

Version: 3 Date: 11 October 2011

Author's response to reviews: see over
Dear Editor,

Find enclosed the revised manuscript. We took all additional requirement into consideration.

1.) We start the method section with an explanation why we do not need approval of an ethics committee. First, we carried out an observational study without any interventions. Furthermore, using general practice based data in the Netherlands approval of an ethics committee is not necessary.

2.) We do not have any acknowledgements, because all participating registration networks have a representative as co-author.

The study has been financed totally with Strategic Research Program grant from the National Institute for Public Health and the Environment.

If there are any questions, please contact me.

Best wishes,

Karin van den Dungen  
Corresponding author

National Institute for Public Health and the Environment (RIVM)  
Bilthoven, the Netherlands  
Karin.van.den.dungen@rivm.nl